Author's response to reviews

Title: TABADO: "Evaluation of a smoking cessation program among Adolescents in Vocational Training Centers": Study protocol.

Authors:

Laetitia Minary (l.minary@chu-nancy.fr)
Hervé Martini Dr (h.martini@chu-nancy.fr)
Nathalie Wirth Dr (n.wirth@chu-nancy.fr)
Francine Thouvenot Dr (francine.thouvenot@wanadoo.fr)
Dovi-Stéphanie Acouetey (Dovi-Stephanie.Acouetey@nancy.inserm.fr)
Yves Martinet Pr (y.martinet@chu-nancy.fr)
Abraham Bohadana Dr (abraham.bohadana@nancy.inserm.fr)
Denis Zmirou-Navier Pr (denis.zmirou@nancy.inserm.fr)
François Alla Pr (f.alla@chu-nancy.fr)

Version: 2 Date: 29 September 2009

Author's response to reviews:

Dear Editor,

Please find enclosed an amended manuscript of "TABADO: Evaluation of a smoking cessation program among Adolescents in Vocational Training Centers: Protocol study," by Minary et al. that we wish to have it evaluated for publication in BioMed Central Public Health.

As requested, the manuscript above keeps the detailed description of the trial and the trial registration number has been cited in the abstract.

We hope this manuscript will be found acceptable for publication in BioMed Central Public Health.

Sincerely,

L.Minary